Author's response to reviews

Title: Rationale, design and implementation protocol of an electronic health record Integrated Clinical Prediction Rule (iCPR) randomized trial in primary care

Authors:

Devin M Mann (dmann@bu.edu)
Joseph L Kannry (joseph.kannry@mssm.edu)
Daniel Edonyabo (daniel.edonyabo@mssm.edu)
Alice C Li (alicecli@gmail.com)
Jacqueline Arciniega (jacqueline.arciniega@nyumc.org)
James Stulman (james.stulman@mssm.edu)
Lucas Romero (lucas.romo@nyumc.org)
Juan Wisnivesky (juan.wisnivesky@mssm.edu)
Rhodes Adler (rhodes.adler@mssm.edu)
Thomas P McGinn (tmcginn@nshs.edu)

Version: 3 Date: 23 July 2011

Author's response to reviews:

The missing CONSORT checklist is now attached.